Author's response to reviews

Title: Effects of an exercise programme for chronically ill and mobility-restricted elderly with structured support by the general practitioner’s practice (HOMEfit) - Study protocol of a randomised controlled trial

Authors:

Timo Hinrichs (timo.hinrichs@rub.de)
Anna Moschny (anna.moschny@rub.de)
Michael Brach (michael.brach@uni-muenster.de)
Stefan Wilm (stefan.wilm@uni-wh.de)
Renate Klaaßen-Mielke (renate.klaassen-mielke@rub.de)
Matthias Trampisch (matthias.trampisch@rub.de)
Petra Platen (petra.platen@rub.de)

Version: 2 Date: 10 December 2011

Author's response to reviews:

Dear Editors

The following formatting changes (marked by blue font) have been made to the manuscript:

1. The abstract has been shortened to meet the given limit of 300 words (pages 2-3).
2. The section ‘Figure legends’ has been shifted (from the end of the document) directly behind the references (page 50).
3. A ‘Trial Status’ section has been included (page 38).
4. We revised the citing of the references. They are now cited in the correct numerical order.
5. A few minor corrections of single words have been made without changing the sense or content (page 32, 34, 35, 37).
6. The format (font and column width) of table 4 (page 53) has been changed.

Best regards
Timo Hinrichs